%%%%%%%%%%%%%%%%%%%%%%%% referenc.tex %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% sample references
% %
% Use this file as a template for your own input.
%
%%%%%%%%%%%%%%%%%%%%%%%% Springer-Verlag %%%%%%%%%%%%%%%%%%%%%%%%%%
%
% BibTeX users please use
% \bibliographystyle{}
% \bibliography{}
%
%% \biblstarthook{In view of the parallel print and (chapter-wise) online publication of your book at \url{www.springerlink.com} it has been decided that -- as a genreral rule --  references should be sorted chapter-wise and placed at the end of the individual chapters. However, upon agreement with your contact at Springer you may list your references in a single seperate chapter at the end of your book. Deactivate the class option \texttt{sectrefs} and the \texttt{thebibliography} environment will be put out as a chapter of its own.\\\indent
%% References may be \textit{cited} in the text either by number (preferred) or by author/year.\footnote{Make sure that all references from the list are cited in the text. Those not cited should be moved to a separate \textit{Further Reading} section or chapter.} The reference list should ideally be \textit{sorted} in alphabetical order -- even if reference numbers are used for the their citation in the text. If there are several works by the same author, the following order should be used: 
%% \begin{enumerate}
%% \item all works by the author alone, ordered chronologically by year of publication
%% \item all works by the author with a coauthor, ordered alphabetically by coauthor
%% \item all works by the author with several coauthors, ordered chronologically by year of publication.
%% \end{enumerate}
%% The \textit{styling} of references\footnote{Always use the standard abbreviation of a journal's name according to the ISSN \textit{List of Title Word Abbreviations}, see \url{http://www.issn.org/en/node/344}} depends on the subject of your book:
%% \begin{itemize}
%% \item The \textit{two} recommended styles for references in books on \textit{mathematical, physical, statistical and computer sciences} are depicted in ~\cite{science-contrib, science-online, science-mono, science-journal, science-DOI} and ~\cite{phys-online, phys-mono, phys-journal, phys-DOI, phys-contrib}.
%% \item Examples of the most commonly used reference style in books on \textit{Psychology, Social Sciences} are~\cite{psysoc-mono, psysoc-online,psysoc-journal, psysoc-contrib, psysoc-DOI}.
%% \item Examples for references in books on \textit{Humanities, Linguistics, Philosophy} are~\cite{humlinphil-journal, humlinphil-contrib, humlinphil-mono, humlinphil-online, humlinphil-DOI}.
%% \item Examples of the basic Springer style used in publications on a wide range of subjects such as \textit{Computer Science, Economics, Engineering, Geosciences, Life Sciences, Medicine, Biomedicine} are ~\cite{basic-contrib, basic-online, basic-journal, basic-DOI, basic-mono}. 
%% \end{itemize}
%% }

\begin{thebibliography}{99.}%

\bibitem{AgAlmouloud} Ag Almouloud S. (1992) L’ordinateur, outil d’aide à l’apprentissage
 de la démonstration et de traitement de données didactiques, Thèse de
 doctorat de l’Université de Rennes 1.

 
\bibitem{Agrawal}Agrawal R., Imielinsky T. et Swami A (1993) Mining association rules between sets of items in large databases, Proc. of the ACM SIGMOD'93.

\bibitem{Amarger} Amarger S., Dubois D. and Prade H. (1991) Imprecise quantifiers and conditional probabilities - in Symbolic and quantitative approaches to uncertainty  (R. KRUSE, P. SIEGEL), Springer-Verlag, 33-37.

  
\bibitem{Atlana} Atlan H. (1986), A tort et à raison, Seuil.
\bibitem{Atlanb}  Atlan H. (2014) Croyances, Paris, Autrement.

  
 \bibitem{Bachelard} Bachelard G. (1967) La Formation de l'esprit scientifique, Paris, 5e édition, Librairie philosophique, J. Vrin.

 \bibitem{Bailleula} Bailleul M. (1994) Analyse statistique
   implicative: variables modales et contribution des
   sujets. Application à la modélisation de l'enseignant dans le
   système didactique, Thèse de l'Université de Rennes 1, juin 1994.
   
 \bibitem{Bailleulb} Bailleul M., Gras R. (1995) L'implication statistique entre variables modales, Mathématique, Informatique et Sciences Humaines, E.H.E.S.S. Paris, n°128, 41-57.

   
\bibitem{Benzecri}  Benzecri, J.P. (1973) L’analyse des données (vol 1), Dunod, Paris. 
\bibitem{Bernard} Bernard J.-M. and Poitrenaud S. (1999) L'analyse implicative bayesienne d'un questionnaire binaire : quasi-implications et treillis de Galois simplifié", Mathématiques, Informatique et Sciences Humaines,  n° 147, 1999, 25-46



\bibitem{Blancharda}  Blanchard J., Kuntz P., Guillet F. and Gras R. (2004) Mesure de la qualité des régles d’association par l’intensité d’implication entropique, Mesures de qualité pour la fouille de données, RNTI-E-1, p. 33-44.
  
\bibitem{Blanchardb} Blanchard J., Guillet F., Briand H. and Gras R. (2005) Ipee: Indice probabiliste d'écart à l'équilibre pour l'évaluation de la qualité des règles, actes Atelier Qualité des Données et des Connaissances, pp. 26-34.
  
\bibitem{Blanchardc} Blanchard J., Guillet F. and Gras R. (2008)  Assessing the interestingness of temporal rules, Statistical Implicative Analysis, R.Gras, E. Suzuki, F.Guillet and F.Spagnolo, Eds, Springer-Verlag, Berlin-Heidelberg, ISBN 978-3-540-78982-6.

  
  
\bibitem{Blanchard} Blanchard J., Guillet F. et Gras R. (2009) Analyse Implicative Séquentielle, Analyse Statistique Implicative, Une méthode d’analyse de données pour la recherche de causalités, dir. R.Gras, eds R.Gras, J.-C. Régnier, Guillet F. Cépaduès, Toulouse, p. 183-194. 


\bibitem{Bodina} Bodin A. (1997) Modèles sous-jacents à l'analyse implicative et outils complémentaires. Prépublication IRMAR. n°97-32.
  
\bibitem{Bodinb} Bodin A. and Gras R. (1999) Analyse du préquestionnaire enseignants avant EVAPM-Terminales, Bulletin n°425 de l'Association des Professeurs de Mathématiques de l'Enseignement Public, 772-786, Paris.
  
  
\bibitem{Couturiera} Couturier R. (2001) Traitement de l’analyse
  statistique implicative dans CHIC, Actes des Journées sur la
   Fouille dans les données par la méthode d’analyse implicative, IUFM
  Caen, p. 33-50.
  
\bibitem{Couturierb} Couturier R, Gras R., et Guillet F. (2004)
  Reducing the number of variables using implicative analysis In
  International Federation of Classification Societies, IFCS 2004,
  Springer Verlag: Classification, Clustering, and Data Mining
  Applications, p. 277--285, ISBN 3-540-22014-3, Chicago, July 2004.
  
\bibitem{Couturierc}Couturier R, Gras R. (2005) CHIC: Traitement de
  données avec l’analyse implicative, Extraction et Gestion des
  Connaissances, Volume I1, RNTI, Cépaduès, Paris, p.679-684, , ISBN
  2.85428.683.9
  
\bibitem{Couturierd} Couturier R. et Ag Almouloud S. (2013) Historique et fonctionnalités de CHIC, L’analyse statistique implicative, Méthode exploratoire et confirmatoire à la recherche de causalités, sous la direction de Gras R., eds Gras R., Régnier J.-C., Marinica C., Guillet F., Cépaduès Editions,, p.313-325.
  
\bibitem{Davida} David J, Guillet F., Gras R. and Briand H. (2006) 
  Conceptual hierarchies matching : an approach based on discovery of
  implication rules between concepts, In Proc. ECAI 2006, 17th
  European Conference on Artificial Intelligence, Riva del Garda,
  Italy, IOS Press
  
\bibitem{Davidb} David J., Guillet F., Gras R. (2017) Alignement
  extensionnel et asymétrique d’ontologies par découverte
  d’implications entre concepts, L’Analyse Statistique Implicative,
  des Sciences dures aux Sciences Humaines et Sociales, R. Gras (dir.), Cépaduès Ed. Toulouse, p. 339-448, ISBN: 978.2.36493.577.8.

\bibitem{Diday} Diday E. (1972) Nouvelles méthodes et nouveaux concepts en classification automatique et reconnaissance des formes,  Thèse d'Etat, Université de Paris VI, 1972.
  

\bibitem{Ehrenberg} Ehrenberg A. (2008) Sciences Humaines, n° 198, nov. 2008. 


\bibitem{Ekeland} Ekeland I. (2002) La complexité, vertiges et promesses, Le Pommier.
  
  
\bibitem{Espagnat} d’Espagnat B. (1981) A la recherche du réel, Le
  regard d’un physicien, Paris.

\bibitem{Fleury} Fleury L. (1996) Extraction de connaissances dasn une base de données pour la gestion de ressources humaines, Thèse d’Université, Université de Nantes, 22 novembre 1996.

  
\bibitem{Foucault} Foucault M. (1966) Les mots et les choses, Éditions
  Gallimard, Paris.

\bibitem{Gaudin} Gaudin F.  (2005) Emergence, complexité et dialectique, Paris, O.Jacob.


  
\bibitem{Grasa}  Gras R. (1976) Recherche d’une taxonomie d’objectifs cognitifs en Mathématiques, I.R.E.M. de Rennes. 

\bibitem{Grasb} Gras R. (1979) Contribution à l'étude expérimentale et à l'analyse de certaines acquisitions cognitives et de certains objectifs didactiques en mathématiques, Thèse d'Etat, Université de Rennes I.

\bibitem{Grasc} Gras R.,  Larher A. (1992) L'implication statistique, une nouvelle méthode d'analyse de données,  Mathématique, Informatique et Sciences Humaines, E.H.E.S.S. Paris, n°120,  1992, 5-31.

\bibitem{Grasd} Gras R., Totohasina A. (1995) Conceptions d'élèves sur la notion de probabilité conditionnelle révélées par une méthode d'analyse des données: Implication - similarité – corrélation, Educational Studies in Mathematics 28(4):337-363.

\bibitem{Grase} Gras R., Ratsimaba-Rajohn H. (1996) Analyse non symétrique de données par l'implication statistique. RAIRO-Recherche Opérationnelle, 30-3, AFCET, Paris, 1996, 217-232.

\bibitem{Grasf} Gras R., Ag Almouloud S., Bailleul M., Larher A., Polo M., Ratsimba-Rajohn et H., Totohasina A. (1996) L’implication statistique, Nouvelle méthode d’analyse de données, La Pensée Sauvage Editions, Grenoble. 

\bibitem{Grasg} Gras R., Briand H., Peter P. (1996) Structuration sets with implication intensity", Proceedings of the International Conference on Ordinal and Symbolic Data Analysis - OSDA 95, E.Diday, Y.Chevallier, Otto Opitz, Eds., Springer, Paris, 147-156.

\bibitem{Grash} Gras R., Briand H., Peter P., Philippé J. (1997) Implicative statistical analysis, Proceedings  of International Congress I.F.C.S., 96, Kobé, Springer-Verlag, Tokyo , 1997, 412-419
Gras R., Kuntz P., Couturier R. et Guillet F., (2001), Une version entropique de l'intensité d'implication pour les corpus volumineux,.Extraction des Connaissances et Apprentissage (ECA), vol. 1, n° 1-2, 69-80. Hermès Science Publication.

\bibitem{Grasi} Gras R., Kuntz P. et Briand H. (2001) Les fondements de l'analyse
statistique implicative et quelques prolongements pour la fouille de
données, Mathématiques et Sciences Humaines, n° 154-155, p 9-29, ISSN
0987 6936.

\bibitem{Grasj} Gras R., Diday E., Kuntz P.,  Couturier R. (2001) Variables sur intervalles et variables-intervalles en analyse statistique implicative, Actes du 8ème Congrès de la Société Francophone de Classification, Université des Antilles-Guyane, Pointe-à-Pitre, 17-21 décembre 2001, pp 166-173.

\bibitem{Grask} Gras R(égis), Guillet F., Gras R(obin). et Philippé J. (2002) Réduction des colonnes d’un tableau de données par quasi-équivalence entre variables, Extraction des connaissances et apprentissage, Hermès, Volume 1, n°4/2001,  p 197-202, ISBN 2-7462-0406-1.

\bibitem{Grasl} Gras R., Kuntz P. et Briand H. (2003) Hiérarchie orientée de règles généralisées en analyse implicative, Extraction des Connaissances et apprentissage, Hermès, p 145-157, ISSN 0992-499X, ISBN 2-7462-0631-5.

\bibitem{Grasm} Gras R., Couturier R.., Blanchard J., Briand H., Kuntz P., Peter P. (2004)  Quelques critères pour une mesure de qualité de règles d’association. Un exemple : l’implication statistique,  Mesures de qualité pour la fouille de données, RNTI-E-1, Cépaduès Editions, p 3-32, I.S.B.N. 2.85428.646.4.

\bibitem{Grasn} Gras R., Kuntz P. et Régnier J.C. (2004)
  Significativité des niveaux d’une hiérarchie orientée en analyse
  statistique implicative, Classification et fouille de données, Ed
  M. Chavent et M.Langlais, RNTI-C-1, Cepaduès, p. 39-50,ISBN
  2.85428.667.7.
  
\bibitem{Graso} Gras R., Kuntz P. (2005) Discovering R-rules with a directed hierarchy, Soft Computing, A Fusion of Foundations, Methodologies and Applications, Volume 1,  p. 46-58, ISSN 1432-7643, Springer Verlag.

\bibitem{Grasp} Gras R., David J., Régnier J.C., Guillet F. (2006) Typicalité et contribution des sujets et des variables supplémentaires en Analyse Statistique Implicative, Extraction des Connaissances ( EGC’06),Volume2,  Cépaduès Editions, p.359-370, ISBN: 2.85428.723.1.

\bibitem{Grasq} Gras R., Suzuki E. et Kuntz P. (2007) Règle et R-règle d’exception en Analyse Statistique Implicative, Nouveaux apports théoriques à l’Analyse Statistique Implicative et Applications, R.Gras, P.Orus, P.Gregori, B.Pinaud, eds,  4èmes Rencontres Internationales d’Analyse Statistique Implicative, Université Jaume I de Castellon 18-21/10/2007, p. 305-313, ISBN 978-84-690-8241-6.

\bibitem{Grasr} Gras R., Kuntz P., Suzuki E. (2007). Une règle
  d’exception en Analyse Statistique Implicative, Extraction des
  Connaissances ( EGC’07),Volume1, RNTI-E-9, Cépaduès Editions, p.87-98.

  
\bibitem{Grass} Gras R., Kuntz P. (2008) An overview of the
  Statistical Implicative, Statistical Implicative Analysis, R.Gras,
  E. Suzuki, F. Guillet and F. Spagnolo, Eds,  Springer-Verlag, Berlin-Heidelberg, ISBN 978-3-540-78982-6.

\bibitem{Grast} Gras R., Régnier J.C. (2009) Qualité d’un graphe implicatif: variance implicative, Analyse Statistique Implicative, Une méthode d'analyse de données pour la recherche de causalités, sous la direction de Régis Gras, réd, invités Régis Gras, Jean-Claude Régnier, Fabrice Guillet, Cépaduès Ed. Toulouse, ISBN : 978.2.85428.8971, p.151-163.

\bibitem{Grast2} Gras R., Couturier R. (2010) Spécificités de
  l'Analyse Statistique Implicative (A.S.I.) par rapport à d'autres
  mesures de qualité de règles d'association, Conference: 5ème colloque international ASI (ASI 5).

  
\bibitem{Grasu} Gras R., Couturier R. (2012) Implication entropique et causalité, L’Analyse Statistique Implicative : de l’exploratoire au confirmatoire, Eds J.C.Régnier, M.Bailleul, R.Gras, Université de Caen, ISBN 978-2-7466-5256-9, p.39-50.

\bibitem{Grasv} Gras R., Lahanier-Reuter D. (2012) Dualité entre espace des variables et espace des sujets, L’Analyse Statistique Implicative: de l’exploratoire au confirmatoire, Eds J.C.Régnier, M.Bailleul R.Gras, Université de Caen, ISBN 978-2-7466-5256-9, p. 19-38.

\bibitem{Grasw} Gras R., Régnier J.C. (2012) Extension de l’analyse statistique implicative au cas continu de l’espace des sujets, L’Analyse Statistique Implicative: de l’exploratoire au confirmatoire, Eds J.C.Régnier, M.Bailleul R.Gras, Université de Caen, ISBN 978-2-7466-5256-9, p. 51-63.

\bibitem{Grasx} Gras R. (2017) Un survol paradigmatique de l'Analyse Statistique Implicative, L’Analyse Statistique Implicative, des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.), Cépaduès Ed. Toulouse, p. 27-40, ISBN : 978.2.36493.577.8.

\bibitem{Grasy} Gras R., Acioly-Régnier N. (2017) Hiérarchie de règles en A.S.I. et conceptualisation, L’Analyse Statistique Implicative, des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.), Cépaduès Ed. Toulouse, p.369-378, ISBN : 978.2.36493.577.8.

\bibitem{Grasz} Gras R., Régnier J.-C. (2017) Extension de l’Analyse Statistique Implicative au cas des variables continues quelconques, L’Analyse Statistique Implicative, des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.), Cépaduès Ed. Toulouse, p. 157-166, ISBN : 978.2.36493.577.8.

\bibitem{Grasaa} Gras R., Kuntz P., Greffard N. (2017) Notion de champ implicatif en Analyse Statistique Implicative, L’Analyse Statistique Implicative, des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.), Cépaduès Ed. Toulouse, p. 167-178, ISBN: 978.2.36493.577.8.

\bibitem{Grasab} Gras R., Couturier R., Gregori P. (2017) Un mariage arrangé entre
l’implication et la confiance?, L’Analyse Statistique Implicative,
des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.),
Cépaduès Ed. Toulouse, p. 195-208, ISBN: 978.2.36493.577.8.

\bibitem{Guillet} Guillet, F., Hamilton, H. J.  (2007) Quality measures in data mining (Vol. 43). Springer.


\bibitem{Jacquard} Jacquard A. (2001) La science à l’usage des non-scientifiques », p.159, 2001.



\bibitem{Kodratoff} Kodratoff Y. (2000) Extraction de connaissances à partir des données
 et des textes, Actes du colloque ASI 2, IUFM de Caen. 

\bibitem{Lallich} Lallich S., Lenca P. and Vaillant B. (2005) Variations autour de l’intensité d’implication, Actes ASI 05, Université de Palerme

 
\bibitem{Lagrange} Lagrange J.B. (1998) Analyse implicative d'un ensemble de variables numériques ; application au traitement d'un questionnaire à réponses modales ordonnées", Revue de Statistique Appliquée., Institut Henri Poincaré, Paris, 71-93.

 
\bibitem{Lahaniera}Lahanier-Reuter D. (1998)Etude de conceptions du hasard : approche épistémologique, didactique et expérimentale en milieu universitaire, Thèse de l'Université de Rennes 1.
 
\bibitem{Lahanierb}Lahanier-Reuter D. (1999) Conceptions du hasard et enseignement des probabilités et statistiques, Éducation et Formation, PUF.

\bibitem{Lahanierc}Lahanier-Reuter D., Gras R., Bailleul M. (2017), Variable nodale et cône implicatif,  L’Analyse Statistique Implicative, des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.), Cépaduès Ed. Toulouse, p. 287-302, ISBN: 978.2.36493.577.8.

\bibitem{Lauvergeon} Lauvergeon A. (2012) La femme qui résiste, Plon.

  
\bibitem{Lebart} Lebart L., Morineau A. and Piron M., Statistique exploratoire multidimensionnelle, Dunod.

\bibitem{Leblond} Leblond J.M. (1996) Aux contraires, nrf essais, Paris, Gallimard.

  
\bibitem{Lencaa} Lenca P., Meyer P., Vaillant P., Picouet P. and
  Lallich S., (2004), Evaluation et analyse multi-critères de qualité
  des règles d’association, Mesures de qualité pour la fouille de
  données, RNTI-E-1,Cépaduès,  p. 219-246.

\bibitem{Lencab} Lenca P., Vaillant B., Meyer P., and Lallich
  S. (2007), Association Rule Interestingness Measures: Experimental
  and Theorical Studies, Guillet F. and Hamilton H.J. eds, Studies in
  Computational Intelligence 43, Springer, p. 51-76.
  
\bibitem{Lent} Lent B., A.N. Swami A.N., et J. Widow J. (1997), Clustering association rules. Proc. of the 13th Int. Conf. on Data Engineering, p. 220-231.

  
\bibitem{Lerman} Lerman, I. C. (1970) Sur l'analyse des données préalable à une classification automatique (proposition d'une nouvelle mesure de similarité). Mathématiques et sciences humaines, 32, 5-15.
  
\bibitem{Lermana} Lerman I.-C., Gras R. and Rostam H. (1981) Elaboration et évaluation d'un indice d'implication pour des données binaires, I et  II, Mathématiques et Sciences Humaines, n°74,, 5-35 and n° 75, 5-47
  
\bibitem{Lermanb} Lerman I.C. (1981) Classification et analyse ordinale des données, Paris, Dunod


\bibitem{Lermanc} Lerman I.C. and Azé J. (2004) Indice probabiliste discriminant (de vraisemblance du lien) d’une règle d’association en cas de «très grosses» données, Mesures de qualité pour la fouille de données, RNTI-E-1, p. 69-94, 2004.


  
\bibitem{Levi-strauss} Lévi-Strauss C. (1967) Structures élémentaires
  de la parenté, De Gruyter Mouton.



  \bibitem{Levy-Leblonda}Lévy-Leblond J.-M. (1981) L'esprit de sel, Fayard.

\bibitem{Levy-Leblondb}Lévy- Leblond J.-M. (1996) Aux contraires,  NRF, Paris

\bibitem{Levy-Leblondc}Lévy-Leblond, J.-M. (2006) La vitesse de l’ombre, Seuil.
  
\bibitem{Loevinger} Loevinger J. (1947), A systematic approach to the
  construction and evaluation of tests of abilities, Psychological Monographs, 61, n° 4.


\bibitem{Marx} Marx K. (1857-1858) Manuscrits de 1857-1858.


\bibitem{Ratsimba} Ratsimba-Rajohn H. (1992) Contribution à l’étude de
  la hiérarchie implicative. Application à l’analyse de la gestion
  didactique des phénomènes d’ostension et de contradiction, Thèse de
  doctorat de l’Université de Rennes 1.

\bibitem{Ndonga}   Ndong L. (2008) La place des concepts de la didactique des
   sciences dans la formation des professeurs de lycée et collège de
   Sciences de la Vie et de la Terre en France et au Gabon. Thèse
   d’Université, Paris 5,

   
\bibitem{Ndongb} Ndong L., Gras R. (2017) Classification des primates et Classification hiérarchique orientée de l’A.S.I., L’Analyse Statistique Implicative, des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.), Cépaduès Ed. Toulouse, p.453-458, ISBN: 978.2.36493.577.8.


\bibitem{Pasquiera} Pasquier D., Gras R., Couturier R. (2010) L'analyse statistique implicative vectorielle ; une application au domaine de la description personnologique en passation répétée, Eds Régnier J.C., Spagnolo F., Di Paola B., Gras R., Actes du Colloque ASI 5, Université de Palerme., p.351-353, ISSN 1592-5137.
\bibitem{Pasquierb} Pasquier D., Gras R. (2013) Analyse Statistique Implicative et psychométrie, Méthode exploratoire et confirmatoire à la recherche de causalités, sous la direction de Gras R., eds Gras R., Régnier J.-C., Marinica C., Guillet F., Cépaduès Editions, 522 pages, ISBN 978.2.36493.056.8, p. 365-378.

\bibitem{Pasquierc} Pasquier D., Gras  R. (2017) De l’intérêt de l’Analyse Statistique Implicative (A.S.I.) pour la recherche exploratoire en psychologie, L’Analyse Statistique Implicative, des Sciences dures aux Sciences Humaines et Sociales, R.Gras (dir.), Cépaduès Ed. Toulouse, p.379-386, ISBN: 978.2.36493.577.8


  
\bibitem{Ramstein} Ramstein G. (2009) Une méthode implicative pour l’analyse des données d’expression de gènes, L’Analyse Statistique Implicative. Méthode exploratoire et confirmatoire à la recherche de causalités, sous la direction de direction de Régis Gras, (Eds.)  R. Gras, J.C. Régnier, C. Marinica, F. Guillet, Cépaduès Ed. Toulouse, p 481-496. 
  
\bibitem{Regniera} Régnier J.-C., Gras R. (2005), Statistique de
  rangs et analyse Statistique Implicative, Revue de Statistique
  Appliquée, LIII, p. 5-38.
  
\bibitem{Regnierb} Régnier J.-C.,  Acioly-Régnier, N. M. (2007) Analyse cohésitive et interprétations des données dans le champ de l’éducation. In Régis Gras et al. Nouveaux apports théoriques à l’analyse statistique implicative et applications. Castellón: Innovació Digital Castelló. ISBN : 978-84-690-8241-6. p.329-343.

\bibitem{Saporta} Saporta G. (1990) Probabilités, Analyse de Données et statistique, Ed. Technip, Paris.

  
\bibitem{Schelling}   Schelling F.-W. (1994) Philosophie de la mythologie, p.361, Paris, J. Millon
  
\bibitem{Seve} Sève L. (2005) Émergence, complexité et dialectique, Paris.


\bibitem{Simon} Simon A. (2001),Outils classificatoires par objets pour l'extraction de connaissances dans des bases de données, Thèse de l'Université de Nancy 1.

\bibitem{Shannon} Shannon C. E. and Weaver W. (1949) The mathematical theory of communication, Univ. of Illinois Press.

  
\bibitem{Shapin} Shapin S. (2014) Une histoire sociale de la vérité, La Découverte, Paris.



\bibitem{Spagnolo}  Spagnolo F., Gras R. (2004) A new approach in Zadeh’s classification : fuzzy implication through statistic implication, NAFIPS 2004, 23nd Conference of the North American Fuzzy Information Processing Society, june 27-30, Banff, AB Canada.

  
\bibitem{Thoma} Thom R. (1983) Paraboles et catastrophes, Champs Sciences. 
\bibitem{Thomb} Thom R. (1993) Prédire n'est pas expliquer, Champs Sciences. 


    
%% % and use \bibitem to create references.
%% %
%% % Use the following syntax and markup for your references if 
%% % the subject of your book is from the field 
%% % "Mathematics, Physics, Statistics, Computer Science"
%% %
%% % Contribution 
%% \bibitem{science-contrib} Broy, M.: Software engineering --- from auxiliary to key technologies. In: Broy, M., Dener, E. (eds.) Software Pioneers, pp. 10-13. Springer, Heidelberg (2002)
%% %
%% % Online Document
%% \bibitem{science-online} Dod, J.: Effective substances. In: The Dictionary of Substances and Their Effects. Royal Society of Chemistry (1999) Available via DIALOG. \\
%% \url{http://www.rsc.org/dose/title of subordinate document. Cited 15 Jan 1999}
%% %
%% % Monograph
%% \bibitem{science-mono} Geddes, K.O., Czapor, S.R., Labahn, G.: Algorithms for Computer Algebra. Kluwer, Boston (1992) 
%% %
%% % Journal article
%% \bibitem{science-journal} Hamburger, C.: Quasimonotonicity, regularity and duality for nonlinear systems of partial differential equations. Ann. Mat. Pura. Appl. \textbf{169}, 321--354 (1995)
%% %
%% % Journal article by DOI
%% \bibitem{science-DOI} Slifka, M.K., Whitton, J.L.: Clinical implications of dysregulated cytokine production. J. Mol. Med. (2000) doi: 10.1007/s001090000086 
%% %
%% \bigskip

%% % Use the following (APS) syntax and markup for your references if 
%% % the subject of your book is from the field 
%% % "Mathematics, Physics, Statistics, Computer Science"
%% %
%% % Online Document
%% \bibitem{phys-online} J. Dod, in \textit{The Dictionary of Substances and Their Effects}, Royal Society of Chemistry. (Available via DIALOG, 1999), 
%% \url{http://www.rsc.org/dose/title of subordinate document. Cited 15 Jan 1999}
%% %
%% % Monograph
%% \bibitem{phys-mono} H. Ibach, H. L\"uth, \textit{Solid-State Physics}, 2nd edn. (Springer, New York, 1996), pp. 45-56 
%% %
%% % Journal article
%% \bibitem{phys-journal} S. Preuss, A. Demchuk Jr., M. Stuke, Appl. Phys. A \textbf{61}
%% %
%% % Journal article by DOI
%% \bibitem{phys-DOI} M.K. Slifka, J.L. Whitton, J. Mol. Med., doi: 10.1007/s001090000086
%% %
%% % Contribution 
%% \bibitem{phys-contrib} S.E. Smith, in \textit{Neuromuscular Junction}, ed. by E. Zaimis. Handbook of Experimental Pharmacology, vol 42 (Springer, Heidelberg, 1976), p. 593
%% %
%% \bigskip
%% %
%% % Use the following syntax and markup for your references if 
%% % the subject of your book is from the field 
%% % "Psychology, Social Sciences"
%% %
%% %
%% % Monograph
%% \bibitem{psysoc-mono} Calfee, R.~C., \& Valencia, R.~R. (1991). \textit{APA guide to preparing manuscripts for journal publication.} Washington, DC: American Psychological Association.
%% %
%% % Online Document
%% \bibitem{psysoc-online} Dod, J. (1999). Effective substances. In: The dictionary of substances and their effects. Royal Society of Chemistry. Available via DIALOG. \\
%% \url{http://www.rsc.org/dose/Effective substances.} Cited 15 Jan 1999.
%% %
%% % Journal article
%% \bibitem{psysoc-journal} Harris, M., Karper, E., Stacks, G., Hoffman, D., DeNiro, R., Cruz, P., et al. (2001). Writing labs and the Hollywood connection. \textit{J Film} Writing, 44(3), 213--245.
%% %
%% % Contribution 
%% \bibitem{psysoc-contrib} O'Neil, J.~M., \& Egan, J. (1992). Men's and women's gender role journeys: Metaphor for healing, transition, and transformation. In B.~R. Wainrig (Ed.), \textit{Gender issues across the life cycle} (pp. 107--123). New York: Springer.
%% %
%% % Journal article by DOI
%% \bibitem{psysoc-DOI}Kreger, M., Brindis, C.D., Manuel, D.M., Sassoubre, L. (2007). Lessons learned in systems change initiatives: benchmarks and indicators. \textit{American Journal of Community Psychology}, doi: 10.1007/s10464-007-9108-14.
%% %
%% %
%% % Use the following syntax and markup for your references if 
%% % the subject of your book is from the field 
%% % "Humanities, Linguistics, Philosophy"
%% %
%% \bigskip
%% %
%% % Journal article
%% \bibitem{humlinphil-journal} Alber John, Daniel C. O'Connell, and Sabine Kowal. 2002. Personal perspective in TV interviews. \textit{Pragmatics} 12:257--271
%% %
%% % Contribution 
%% \bibitem{humlinphil-contrib} Cameron, Deborah. 1997. Theoretical debates in feminist linguistics: Questions of sex and gender. In \textit{Gender and discourse}, ed. Ruth Wodak, 99--119. London: Sage Publications.
%% %
%% % Monograph
%% \bibitem{humlinphil-mono} Cameron, Deborah. 1985. \textit{Feminism and linguistic theory.} New York: St. Martin's Press.
%% %
%% % Online Document
%% \bibitem{humlinphil-online} Dod, Jake. 1999. Effective substances. In: The dictionary of substances and their effects. Royal Society of Chemistry. Available via DIALOG. \\
%% http://www.rsc.org/dose/title of subordinate document. Cited 15 Jan 1999
%% %
%% % Journal article by DOI
%% \bibitem{humlinphil-DOI} Suleiman, Camelia, Daniel C. O'Connell, and Sabine Kowal. 2002. `If you and I, if we, in this later day, lose that sacred fire...': Perspective in political interviews. \textit{Journal of Psycholinguistic Research}. doi: 10.1023/A:1015592129296.
%% %
%% %
%% %
%% \bigskip
%% %
%% %
%% % Use the following syntax and markup for your references if 
%% % the subject of your book is from the field 
%% % "Computer Science, Economics, Engineering, Geosciences, Life Sciences"
%% %
%% %
%% % Contribution 
%% \bibitem{basic-contrib} Brown B, Aaron M (2001) The politics of nature. In: Smith J (ed) The rise of modern genomics, 3rd edn. Wiley, New York 
%% %
%% % Online Document
%% \bibitem{basic-online} Dod J (1999) Effective Substances. In: The dictionary of substances and their effects. Royal Society of Chemistry. Available via DIALOG. \\
%% \url{http://www.rsc.org/dose/title of subordinate document. Cited 15 Jan 1999}
%% %
%% % Journal article by DOI
%% \bibitem{basic-DOI} Slifka MK, Whitton JL (2000) Clinical implications of dysregulated cytokine production. J Mol Med, doi: 10.1007/s001090000086
%% %
%% % Journal article
%% \bibitem{basic-journal} Smith J, Jones M Jr, Houghton L et al (1999) Future of health insurance. N Engl J Med 965:325--329
%% %
%% % Monograph
%% \bibitem{basic-mono} South J, Blass B (2001) The future of modern genomics. Blackwell, London 
%% %
\end{thebibliography}
